Author's response to reviews

Title: Characterization of early host responses in adults with dengue disease

Authors:

Thomas Tolfvenstam (thomas.tolfvenstam@ki.se)  
Anna Lindblom (anna.lindblom@ki.se)  
Mark J Schreiber (mark.schreiber@novartis.com)  
Ling Ling (lingl1@gis.a-star.edu.sg)  
Angelia Chow (angelia.chow@duke-nus.edu.sg)  
Eng Eong Ooi (engeong.ooi@duke-nus.edu.sg)  
Martin L Hibberd (hibberdml@gis.a-star.edu.sg)

Version: 2 Date: 25 January 2011

Author's response to reviews:

The manuscript has been updated with information that the study conformed to the declaration of Helsinki and with details on the local ethics review board that approved the study. In addition, the manuscript now contain the information that the study participants gave their written consent to participate in the study.